

# **Geophysical Research Letters**®

# **RESEARCH LETTER**

10.1029/2021GL093858

# **Key Points:**

- We tested the hypothesis that tree groundwater uptake is related to species identity
- Both rooting depths and the ratio of root to water table depths were significant phylogenetic signals
- Isotopic evidence of groundwater uptake showed a significant binary phylogenetic signal

# Supporting Information:

Supporting Information may be found in the online version of this article.

#### Correspondence to:

J. Evaristo and J. Knighton, j.evaristo@uu.nl; james.knighton@uconn.edu

#### Citation:

Knighton, J., Fricke, E., Evaristo, J., de Boer, H. J., & Wassen, M. J. (2021). Phylogenetic underpinning of groundwater use by trees. *Geophysical Research Letters*, *48*, e2021GL093858. https://doi.org/10.1029/2021GL093858

Received 13 APR 2021 Accepted 23 AUG 2021

 $\ensuremath{\textcircled{}}$  © 2021. The Authors.

This is an open access article under the terms of the Creative Commons Attribution-NonCommercial-NoDerivs License, which permits use and distribution in any medium, provided the original work is properly cited, the use is non-commercial and no modifications or adaptations are made.

# Phylogenetic Underpinning of Groundwater Use by Trees

James Knighton<sup>1</sup>, Evan Fricke<sup>2</sup>, Jaivime Evaristo<sup>3</sup>, Hugo Jan de Boer<sup>3</sup>, and Martin Joseph Wassen<sup>3</sup>

<sup>1</sup>Department of Natural Resources and the Environment, University of Connecticut, Storrs, CT, USA, <sup>2</sup>Department of Biosciences, Rice University, Houston, TX, USA, <sup>3</sup>Copernicus Institute of Sustainable Development, Utrecht University, Utrecht, Netherlands

**Abstract** Root water uptake (RWU) strategies shape climate-vegetation feedbacks and ecosystem productivity. A fundamental relationship between RWU strategies and evolutionary histories (phylogeny) of trees, however, remains poorly understood. Establishing a phylogenetic basis for tree RWU, particularly groundwater use, could improve their representation in terrestrial biosphere models (TBMs) that are crucial for understanding hydrologic and ecosystem responses to climate perturbations. We explored possible phylogenetic bases for tree RWU using two independent data sets: (a) observed root and local groundwater depths representing 502 tree species, and (b) groundwater, soil, and xylem water isotopic evidence for groundwater uptake representing 412 species. Maximum rooting depths ( $RD_{MAX}$ ), the ratio between  $RD_{MAX}$  and mean water table depth (WT) ( $RD_{MAX}$ /WT), and isotopic evidence of groundwater uptake showed significant phylogenetic signals, suggesting that tree RWU strategies are more similar among closely related species. Our findings may be used to parameterize species-level RWU in TBMs, particularly for data-poor regions.

**Plain Language Summary** Transpiration dominates terrestrial evapotranspiration, strongly influencing the movement of water, carbon, nutrients, and pollutants in the landscape. Despite evidence for global variations in water uptake strategies across tree species, the models that simulate water and solute movement through soils and ecosystems mostly neglect this complexity. This is in part because we lack empirical studies to parameterize species-level variations in rooting depth in model code. Our research demonstrates via two independent data sets (observed rooting depths and isotopic evidence of groundwater uptake) that rooting depth has greater degrees of similarity in more closely related species than for distantly related species. Overall, our study finds that phylogenetic relationships of trees could provide reasonable estimates of tree groundwater use, improving global-scale ecosystem models in the absence of empirical studies.

# 1. Introduction

Transpiration mediates the transfer of water and energy between the land surface and the atmosphere (Fan et al., 2017; Good et al., 2015), impacting the regulation of climate and distribution of vegetation over land (Beerling & Berner, 2005). Trees play a particularly relevant role in water and energy transfer because they exhibit a wide range of maximum rooting depths ( $RD_{MAX}$ ). This suggests that trees may be able to access groundwater that is not immediately available for evaporation. Root water uptake (RWU) of local soil moisture and groundwater by trees thereby influences the partitioning of precipitation between surface runoff (immediate streamflow), water infiltrating into forest soils (i.e., catchment stored water) and transpiration. Establishing a possible environmental and physiological basis for  $RD_{MAX}$  and RWU patterns is specifically important because current ecohydrological models at catchment and continental scales tend to oversimplify RWU strategies (Warren et al., 2015). These simplifications in turn limit model veracity and utility for predicting future global change (Fan et al., 2019; Ferguson et al., 2016; Kennedy et al., 2019; Kleidon & Heimann, 1998).

While the evolution of deep roots and their access to groundwater have been associated with major changes in global hydrological and biogeochemical cycles (Beerling & Berner, 2005; Beerling et al., 1998; Ibarra et al., 2019), we cannot yet adequately explain all variations in observed rooting strategies and their ecosystem function (Pierret et al., 2016). Specifically, the degree to which vegetation connects the water table to the atmosphere via RWU of groundwater, as opposed to uptake and transpiration of moisture held under tension in smaller soil pores (Fan, 2015) remains unclear. On the one hand, some trees may gain an energy advantage either by relying on recently infiltrated precipitation in shallow soils or via hydraulic redistribution (Hafner et al., 2017; Sun et al., 2018) of deeper water from neighboring species. On the other hand, trees that invest in the growth and maintenance of deeper rooting systems to access more stable water sources may gain a competitive advantage during periods of drought. A conceptual model of rooting strategies and empirical observations is presented in Figure S1.

A proliferation of studies over the last three decades have used isotopic observations of subsurface waters and tree xylem to provide evidence that some trees are facultative consumers of groundwater (Balugani et al., 2017; Knighton, Conneely, & Walter, 2019; Miller et al., 2010) and others obligate consumers of soil moisture held under tension (Bowling et al., 2017; Evaristo & McDonnell, 2017; Gaines et al., 2016). While local isotopic and sap flux studies have revealed complex RWU strategies in various forest ecosystems (Brum et al., 2019; De Deurwaerder et al., 2018; Gaines et al., 2016; Knighton, Souter-Kline, et al., 2019; Miller et al., 2010; Volkmann et al., 2016), it remains unclear if these empirical observations are climate- and ecosystem-transferable. Given that water is often a limiting resource for plant growth, we would anticipate that  $RD_{MAX}$  and the ratio between  $RD_{MAX}$  and mean annual water table depth (WT),  $RD_{MAX}/WT$ , are traits that relate to RWU strategy and are therefore similar in closely related species.

The environmental and physiological controls on RD<sub>MAX</sub> and RWU in higher plants are poorly known for several related reasons. One body of knowledge suggests that the realized niche for RWU that trees occupy is determined mainly by local environmental conditions (hereafter locality hypothesis), such as soil moisture and topographic positions, regardless of species and life history (Canadell et al., 1996; Evaristo & McDonnell, 2017; Fan et al., 2017, 2019; Hodge, 2004; Schenk & Jackson, 2005). Under this conceptual model, plants are simplified to one monolithic hydrologic functional strategy for RWU that responds to environmental conditions to optimize the water uptake of individual trees. Another body of knowledge suggests that the realized niches for RWU are determined by hydrological niche partitioning in response to neighboring tree species (hereafter niche segregation hypothesis). Niche segregation is the partitioning of subsurface water as a resource through different RWU strategies at sufficiently small spatial scales (Silvertown et al., 2015). The deployment of these RWU strategies is determined mainly by species-level competition for the same amount of available water (Brum et al., 2019; Cabal et al., 2020; Gaines et al., 2016; Magh et al., 2020; Volkmann et al., 2016) or complementary hydraulic redistribution that increases total stand productivity (Grossiord et al., 2014; Hafner et al., 2017; Sun et al., 2018). Niche segregation may be the outcome of local soil moisture and topography as well as neighboring species and is a refinement of the locality hypothesis. Under this conceptual model, the RWU strategies of individual species cannot be separated from that of the stand (Grossiord, 2020; Grossiord et al., 2014; Knighton, Singh, & Evaristo, 2020).

A third body of knowledge suggests that RWU strategies depend on species identity, with closely related trees exhibiting similar strategies (hereafter *phylogenetics hypothesis*). For example, studies have shown consistent evidence for changes in the sources of RWU used by gymnosperms throughout the growing season (Andrews et al., 2012; Berkelhammer et al., 2020; Mackay et al., 2020) and gymnosperms using different water sources from those of neighboring angiosperms (Knighton, Souter-Kline, et al., 2019; Link et al., 2014; Mackay et al., 2020; Tetzlaff et al., 2021). These studies hint at a fundamental, evolutionary basis for commonalities in  $RD_{MAX}$  and RWU strategies. Phylogenetic signals, which describe greater degrees of similarity in more closely related species, could provide a broader context for interpretation of these empirical studies and explain the degree to which taxonomy drives rooting patterns in the presence of external variables such as climate (W. Chen et al., 2013; Kitajima et al., 2010). Root traits have shown consistency among taxonomic classes (Comas & Eissenstat, 2009), as well as significant phylogenetic signals (McCormack et al., 2020; Valverde-Barrantes et al., 2017), though it is unclear if the similarity in traits translates into similar RWU strategy.

We examine tree RWU strategies in the context of a phylogenetic model to address the following hypotheses. H1: there is a significant phylogenetic signal in observed maximum rooting depths, H2: there is a significant phylogenetic signal in the ratio of the maximum rooting depth to the mean annual water table, and H3: there is a significant phylogenetic signal in xylem isotopic evidence of groundwater uptake. We achieve this objective by analyzing two independent global data sets: (a) observed maximum rooting depths ( $RD_{MAX}$ )



and mean annual water table depths (WT) (Fan et al., 2017; Iversen et al., 2017), and (b) groundwater, soil, and xylem water isotopic evidence of groundwater use (Evaristo & McDonnell, 2017).

# 2. Materials, Methods, and Data

#### 2.1. Plant Trait and Hydrologic Data Sets and Data Availability

We analyzed two global data sets that aggregate individual measurements of maximum plant rooting depth ( $RD_{MAX}$ ) and local mean water table depth from a variety of studies (Fan et al., 2017; Iversen et al., 2017). Duplicate records between the databases were removed. We filtered the composite database to retain only records where (a) the study was conducted in a monoculture stand (i.e., only one species was listed in a given study plot), (b) "form/stature" contained the string "tree," (c) records identified trees to species level. This filtering produced a composite database of 1,170 measurements of tree rooting depths covering 641 species. The database contained 304 records covering 268 species with information on both rooting and water table depths.

We examined a second data set based on a meta-analysis that aggregates measurements of soil, groundwater, and tree xylem water isotopic compositions (<sup>2</sup>H, <sup>18</sup>O) collected across ecohydrological studies (Evaristo & McDonnell, 2017) containing 531 observations and 414 unique species. We appended to this database 42 new records published after the original database (see Table S1), increasing the sample size to 573 records and 454 species. The data set includes analysis estimating the proportion of xylem samples for species within each study that presents evidence of groundwater uptake. That is, the number of xylem samples out of a universe of plant samples reported to have groundwater contribution to xylem water. We note that each observation represents a unique study that consists of multiple water isotopic samples of each species.

Each data set (Evaristo & McDonnell, 2017; Fan et al., 2017; Iversen et al., 2017) was validated against The Plant List (TPL), a comprehensive database of accepted plant names (Kalwij, 2012). Records with Latin names that failed to match standardized names in TPL were discarded from analysis ( $n_{\text{RDMAX}} = 139$  species;  $n_{\text{RD/WT}} = 108$  species;  $n_{\text{isotope}} = 2$  species). The databases used for analysis contained 502, 160, and 412 species for analysis of maximum rooting depth (RD<sub>MAX</sub>), and RD<sub>MAX</sub>/WT, and isotopic evidence for groundwater uptake, respectively. The databases contain only 53 species with information on rooting depth, water table depth, and isotopic evidence for groundwater uptake, limiting opportunities for comparisons across the data sets. The geographic distribution of database records is presented in Figure S2.

#### 2.2. Establishment of Phylogenetic Relationships for Trees

To define the evolutionary relationships among organisms, we developed phylogenetic dendrograms of both tree databases with V.Phylomaker (Jin & Qian, 2019) using TPL standardized family, genus, and species names within the R scripting environment. Dendrogram structure was established with algorithm Scenario 3, which determines where a new genus tip is bound with consideration for branch length (Jin & Qian, 2019). All subsequent phylogenetic analyses were based on these established dendrograms. R scripts used in phylogenetic analysis are available (https://github.com/jknigh0813/Phylo\_GW).

# 2.3. Regression Analysis of Tree Rooting and Water Table Depths

We compared the spatial distributions of  $RD_{MAX}$ , local water table depth (WT), and  $RD_{MAX}/WT$  for both angiosperms and gymnosperms. We tested for significant differences in the median latitudes of trees exhibiting isotopic evidence of soil water and groundwater uptake separately for angiosperms and gymnosperms with two-sample Kolmogorov-Smirnov tests. With this and all subsequent hypothesis tests, we evaluated and discuss sensitivity at thresholds for Type 1 errors,  $\alpha$ , of 0.1, 0.05, and 0.01.

We examined the relationship between log-transformed  $RD_{MAX}$  and WT with both standard major axis (SMA) and phylogenetic least squares (PLS) regression. SMA assumes each database observation is independent whereas PLS accounts for the expected covariance structure of regression residuals based on phylogenetic distance and therefore eliminates the assumption that all records are independent. PLS regressions based on phylogenetic dendrograms require single trait values per species. For each species, we

computed the median of  $\text{RD}_{MAX}$  and WT where the database contained multiple records for one species. For each regression, we present the best fit linear regression,  $\ln(\text{RD}) = \beta \times \ln(\text{WT}) + \beta_0$ , 95% confidence intervals on  $\beta$ , and the adjusted  $R^2$ . For PLS, we also present Pagel's  $\lambda_p$ , a measure of the phylogenetic signal strength of regression residuals (Pagel, 1999) (i.e., the degree to which closely related trees show similar deviations from the proposed linear model). We subsequently fitted SMA and PLS to all trees and then separately for angiosperms and gymnosperms. We tested for significant differences in the regression slopes of angiosperms and gymnosperms for both SMA and PLS (Warton et al., 2006).

#### 2.4. Phylogenetic Analysis of Isotopic and Root Trait Evidence of Groundwater Uptake

We evaluated the hypotheses that  $\text{RD}_{MAX}$  (n = 502 species) and  $\text{RD}_{MAX}$ /WT (n = 160 species) (Fan et al., 2017; Iversen et al., 2017) show significant phylogenetic signals (i.e., traits among closely related tree species are significantly more similar than for distantly related species). For each test, where multiple values of  $\text{RD}_{MAX}$  or  $\text{RD}_{MAX}$ /WT were available for a single species we used the median value. The significance of the phylogenetic signals of these plant traits were estimated with both Blomberg's *K* (Blomberg et al., 2003) and Pagel's  $\lambda_p$  (Pagel, 1999). Each test involves a test statistic that can be interpreted as a scaled measure of the strength of a phylogenetic signal. Blomberg's *K* values span ( $0, \infty$ ) where 0 indicates no structure and 1 indicates that the covariance of a trait is proportional to the shared history between species. Pagel's  $\lambda_p$  spans the range of 0–1, where larger values similarly indicate stronger phylogenetic signals. If disagreement occurred between the two tests, we accepted the results of Pagel's  $\lambda_p$  as prior studies have demonstrated that  $\lambda_p$  may be a more robust test than *K* and provides more stable results under the assumption of a Brownian model of evolution (i.e., trait values vary along phylogenesis according to a random walk with mean of 0) (Molina-Venegas & Rodríguez, 2017; Münkemüller et al., 2012).

We evaluated the hypothesis that groundwater uptake emerged from a Brownian model of evolution with n = 412 species (Evaristo & McDonnell, 2017). Each species was assigned a binary variable (0—no groundwater; 1—groundwater) where the operational definition of "groundwater" followed that used by the source research. Where both presence and absence of evidence for groundwater uptake existed (i.e., variations in water sources through time or between individuals), we assigned a value of 1 given that groundwater uptake may be both a consequence of trait variations and environmental conditions. Given that cryogenic vacuum extraction (CVE) may bias stem water <sup>2</sup>H measurements (Y. Chen et al., 2020), the analysis was repeated excluding all records based on CVE (a) using only <sup>2</sup>H in source water identification and (b) analysis based on <sup>2</sup>H and <sup>18</sup>O (see Supporting Information S1). The significance of binary traits was estimated with the *D* statistic (Fritz & Purvis, 2010). *D* values span ( $-\infty$ ,  $\infty$ ), where *D* of 0 indicates Brownian evolution, 1 indicates random trait dispersal, and values above 1 indicate an over dispersed trait.

#### 3. Results

#### 3.1. Geographic Distribution of Rooting Strategies

We observed latitudinal gradients in rooting depth that appear to follow trends in water table depth (WT; Figures 1a and 1b). Gymnosperms generally have shallower  $RD_{MAX}$  than angiosperms (Figure 1a) and occur at higher latitudes in regions with shallower WT (Figure 1b). Across all latitudes, trees exist with  $RD_{MAX}$  both shallower and deeper than the local WT (Figure 1c). The latitudinal distributions of soil water and groundwater use as determined by isotopic studies (Figure 1d) were not significantly different in either angiosperms (*p*-value = 0.197) or gymnosperms (*p*-value = 0.237) as determined by two-sample KS tests.

#### 3.2. Regression Analysis of Rooting and Water Table Depths

SMA regression of log-transformed RD<sub>MAX</sub> against WT suggested that RD<sub>MAX</sub> is strongly correlated with WT across all trees ( $\beta_{SMA} = 0.919$ , 95% confidence interval [0.862, 0.980]; Figure 2a). PLS regression, which accounts for the expected covariance structure of residuals stemming from phylogeny, suggested the relationship between RD<sub>MAX</sub> and WT is somewhat weaker than implied by SMA ( $\beta_{PLS} = 0.658$ , 95% confidence interval [0.567, 0.748]; Figure 2b). These differences are likely attributable to the uneven representation of certain clades in the underlying data set where the structure of residuals exhibited a significant





**Figure 1.** Relationships between latitude and  $RD_{MAX}$  (a), WT (b),  $RD_{MAX}$ /WT (c). Latitudinal distribution of isotopic studies showing soil water and groundwater use in angiosperms and gymnosperms (d).

phylogenetic signal ( $\lambda_p = 0.443$ , *p*-value = 0.000; Figure 2b). Out of 294 individual trees (where both RD<sub>MAX</sub> and WT measurements were available), 105 (35.7%) exhibited RD<sub>MAX</sub>/WT  $\geq 0.98$ . Analysis that clustered vegetation by taxonomic groupings (family) demonstrated that local groundwater depth is not a substantial predictor of RD<sub>MAX</sub> for all tree families, where only 23.4% (11 of 47) exhibited a median RD<sub>MAX</sub>/WT  $\geq 0.98$  (Figure S3). SMA regression suggested no significant differences between gymnosperms and angiosperms (Figure 2a; *p*-value = 0.875). In contrast, PLS analysis identified significant differences (Figure 2b; *p*-value = 0.002) that suggest gymnosperm rooting depths more closely align with local WT ( $\beta_{PLS} = 0.832$ , 95% confidence interval [0.657, 1.003]) than the case is with angiosperms ( $\beta_{PLS} = 0.639$ , 95% confidence interval [0.539, 0.746]).



**Figure 2.** Standard major axis regression (a) and phylogenetic least squares regression (b) for  $\ln(WT)$  and  $\ln(RD_{MAX})$  fit to all data (black), gymnosperms (green), and angiosperms (gray).





**Figure 3.** Global database of angiosperm (gray) and gymnosperm (green) trees showing the absolute value of latitude of the original study (red and blue ring), stable isotopic evidence for groundwater uptake (dark blue), and soil water uptake (light blue), the ratio of rooting depth to water table depth ( $RD_{MAX}$ /WT), and  $RD_{MAX}$ . Colored dots on the inner ring indicate tree family.

# 3.3. Phylogenetic Signal of Rooting Depth and Groundwater Uptake

The data sets describing  $RD_{MAX}$ ,  $RD_{MAX}/WT$  (Fan et al., 2017; Iversen et al., 2017), isotopic evidence of groundwater uptake (Evaristo & McDonnell, 2017), and the absolute value of latitude (i.e., distance from the equator) are arranged on a phylogenetic dendrogram (Figure 3). Not all tree species are represented in these data sets, though we demonstrate that these data sets capture broad variations in rooting depth and that isotopic evidence for groundwater uptake occurs across a range of evolutionary history and latitudes (Figure 3).

RD<sub>MAX</sub> exhibited a significant phylogenetic signal at the  $\alpha \le 0.05$  threshold across 502 tree species as estimated by Pagel's  $\lambda_p$  ( $\lambda_p = 0.272$ , *p*-value = 0.014), and at the  $\alpha \le 0.1$  threshold Blomberg's K ( $K_B = 0.046$ , *p*-value = 0.092; Figure 3). Maximum root depth relative to the local mean water table, RD<sub>MAX</sub>/WT, within 160 tree species was significantly related to phylogenetic structure at the  $\alpha \le 0.01$  threshold with Pagel's  $\lambda_p$  ( $\lambda_p = 0.156$ , *p*-value = 0.001), but not with Blomberg's K ( $K_B = 0.020$ , *p*-value = 0.175; (Figure 3). As previously discussed, we accept the interpretation provided by  $\lambda_p$  based on the result of critical reviews of both statistics for the identification of phylogenetic signals (Molina-Venegas & Rodríguez, 2017; Münkemüller et al., 2012). The  $\lambda_p$  values for RD<sub>MAX</sub> (0.272) and RD<sub>MAX</sub>/WT (0.156) indicated that both phylogenetic signals are weaker than would be expected under a model of perfect Brownian evolution ( $\lambda_p = 1$ ), but significantly greater than 0 (no signal).

Isotopic groundwater and soil water signatures within the xylem water of 412 tree species (Evaristo & McDonnell, 2017) yielded similar results. The *D* statistic of 0.804 indicates that isotopic evidence of groundwater use was significantly different from both 0 (*p*-value = 0.000) and 1 (i.e., random occurrence, *p*-value = 0.000). Similar to the interpretation of  $\lambda_p$  for RD<sub>MAX</sub> and RD<sub>MAX</sub>/WT, the *D* statistic indicates a significant phylogenetic signal, but not as strong as would be expected under a purely Brownian model.

We repeated this phylogenetic analysis excluding all records where water extractions were performed with CVE and analysis using only <sup>2</sup>H. This subset produced similar results (n = 266 species, D = 0.624, *p*-value = 0.000). No significant signal was identified when both <sup>2</sup>H and dual isotope analysis records were discarded (n = 117 species, D = 0.904, *p*-value = 0.203) (see Supporting Information S1).

# 4. Discussion and Conclusions

# 4.1. Locality, Niche, and Phylogenetic Hypotheses for RWU

Prior analyses of observations from diverse climates have provided evidence for a correlation between plant rooting depths and environmental conditions (Barbeta & Peñuelas, 2017; Canadell et al., 1996; Evaristo & McDonnell, 2017; Fan et al., 2017; McCormack et al., 2020, 2020; Schenk & Jackson, 2005). Our SMA regression analysis of the RD<sub>MAX</sub> database suggested that RWU strategies were explained well by the locality hypothesis where RD<sub>MAX</sub> was significantly linearly correlated with WT (Figure 2a). We argue that the SMA regression result (paralleling the methodologies of prior studies) is flawed and occurs only because the underlying statistical test incorrectly assumes independence among database records. A substantial fraction of this database is comprised of several vegetation families that cluster near RD<sub>MAX</sub>/WT = 1 (*Myrtaceae* [n = 16], *Tamaricaceae* [n = 8], and *Pinaceae* [n = 72]; Figure S4). PLS regression analysis indicated that the residuals of the linear regression exhibited a significant phylogenetic signal ( $\lambda_p = 0.443$ ; Figure 2b), and therefore violate the assumptions of SMA regression. In contrast, the PLS regression suggested that rooting depths were only approximately equal to water table depths in gymnosperms (Figure 2b).

Prior studies provide corroborating evidence for variations in RWU between gymnosperms and angiosperms. Meta-analysis of long-term research catchments across five northern/cold regions demonstrated that angiosperms exhibited greater evidence of isotopic overlap with enriched shallow soil moisture whereas gymnosperm xylem water deviated from measured end members (Tetzlaff et al., 2021). Global data indicates that angiosperms invest more in fine roots than gymnosperms (C. Wang et al., 2019), possibly driving variations in RWU. The rooting depth (Fan et al., 2017; Iversen et al., 2017) and isotopic (Evaristo & McDonnell, 2017) data sets exhibited some disagreement on the relative groundwater use by angiosperms and gymnosperms. Gymnosperms, specifically *Pinaceae*, exhibited more consistent evidence of root growth into the saturated zone than angiosperms (Figures 2b and S3) potentially indicating more groundwater use. The global isotopic data set demonstrated less difference between these clades where 36.3% of gymnosperm species exhibited evidence of groundwater uptake versus 36.1% of angiosperms. Notwithstanding, a meta-analysis has shown that the proportion of groundwater contribution to xylem water mixture is greater in angiosperms than in gymnosperms (Evaristo & McDonnell, 2017). These results suggest that more complex models based on phylogenetic dendrograms (Figure 3) are needed to explain variations in tree water use that occur within the broad clades of angiosperms and gymnosperms (Figure 2).

Isotopic data and  $\text{RD}_{MAX}/WT$  showed high variance among angiosperms (Figures 2 and 3), indicating the evolution of diverse water use strategies, including some species that grow deep roots but do not extend to groundwater (Figure 2). Root access to soil water is necessary to avoid xylem embolism and death. Deep roots not reaching groundwater may reflect the depth at which water uptake requires the least energy for the majority of the year (Brantley et al., 2017). Alternately, trees may grow different rooting systems because they vary in the need for energetically available water volumes related to variations in above-ground height (Trugman et al., 2021), stomatal regulation strategies (Klein, 2014), stem resistance to cavitation (Urli et al., 2013), or soil water potential thresholds for root uptake of moisture (Brantley et al., 2017). High variance in  $\text{RD}_{MAX}/WT$  may also reflect ecosystem pressures. Investment in dense shallow rooting systems may allow trees to outcompete neighbors for recent precipitation, whereas deeper roots can potentially sustain growth during periods of drought (Silvertown et al., 2015). Trees with shallow rooting systems may have

evolved to rely on the redistribution of deeper water sources by neighboring species (Hafner et al., 2017; Sun et al., 2018).

The phylogenetic signals of  $RD_{MAX}/WT$  and isotopic evidence for groundwater uptake were significant, but likely shaped by both species identity and environmental conditions (Figure 3). Xylem isotopic measurements collected across mixed-species temperate forests suggest that the seasonal origin of water stored in trees was consistent within each tree species sampled, but differed across species (Allen et al., 2019), possibly indicating the role of species identity in shaping water uptake. Similarly, ecohydrological model calibration to continuous streamflow records demonstrated consistency in derived rooting parameters among monoculture catchments of the same species (Knighton, Singh, & Evaristo, 2020). This result is supported by prior evidence for phylogenetic signals in root traits that may carry implications for RWU (Comas & Eissenstat, 2009; McCormack et al., 2020; Valverde-Barrantes et al., 2017). Some ecosystems show evidence of seasonal oscillations between niche segregation and direct competition driven by seasonal soil water limitation (Andrews et al., 2012; De Deurwaerder et al., 2018; Kulmatiski et al., 2020; McCormack et al., 2020; Rodríguez-Robles et al., 2020), suggesting higher-order interactions between the phylogenetic, niche, and locality hypotheses. In a mixed species forests, co-occurring trees may rely on similar water sources during dry days, but exhibit divergent RWU depths in response to recent precipitation (Grossiord et al., 2017; Liu et al., 2019; Volkmann et al., 2016). In a temperate beech-hemlock catchment, beech trees consistently used older soil water (and therefore more likely tightly bound shallow soil water) than hemlock trees (Knighton, Souter-Kline, et al., 2019). The age of water used by both species varied by season and topographic position within the forest, though the relative differences in xylem water age remained, demonstrating effects of both environmental conditions and species identity on RWU.

#### 4.2. Implications for Improving RWU Representation in Terrestrial Biosphere Models

Climate change is expected to alter the composition of forest ecosystems. Thus, without an adequate representation of the nuanced patterns that exist in RWU across phylogenetic clades, current generation models will fail to capture the dynamics of climate-driven perturbations to ecosystem composition and downstream hydrological impacts. Terrestrial Biosphere Models (TBMs) and hydrological models frequently characterize vegetation with presumed functional groups (e.g., coniferous vs. deciduous or broadleaf vs. needleleaf forest). Our research demonstrates that strategies for water uptake vary considerably within angiosperms (Figures 2 and 3) and therefore heterogeneity in functional groups often described in models. Phylogenetic dendrograms could potentially support a refinement of functional group definitions that would more accurately represent variations in soil water and plant interactions.

Parameterization of RWU for individual species has largely depended on resource intensive measurements of internal tree hydraulics (Yang et al., 2013), xylem water isotopic measurements (Knighton, Kuppel, et al., 2020), soil moisture profiles (Hupet et al., 2003), or spatially integrated measures of latent heat fluxes (Sulis et al., 2019; P. Wang et al., 2018) and stream discharge (Knighton, Singh, & Evaristo, 2020). Recent calls for ecosystem model progress suggest that future research explores parameterizing RWU by assuming a locality hypothesis where critical zone moisture content drives subsurface root growth and subsequent uptake (Fan et al., 2019). However, our re-analysis of global plant rooting depths yielded a contrasting interpretation (Figure 2b).

Our hypothesis of a phylogenetic framing of community RWU patterns could add nuance to conceptual and numerical models of RWU and possibly improve the representation of plant hydraulic strategies in TBMs. Traits of unstudied species can be estimated with rich trait databases and phylogenetic dendrograms, provided the traits are significant phylogenetic signals (Debastiani et al., 2021; Guénard et al., 2013; Penone et al., 2014). The phylogenetic dendrograms established in this research may support accurate estimation of  $RD_{MAX}$ ,  $RD_{MAX}$ /WT, and groundwater uptake for unstudied species and therefore aid model parameterization of vegetation where empirical studies are lacking. Further research is needed to test the viability of phylogenetic parameter estimation for TBMs and hydrological models.

# Data Availability Statement

All data used in this study are previously published (Evaristo & McDonnell, 2017; Fan et al., 2017; Iversen et al., 2017).

#### Acknowledgments

This work was supported by the National Socio-Environmental Synthesis Center under funding received from the National Science Foundation DBI-1639145.

# References

- Allen, S. T., Kirchner, J. W., Braun, S., Siegwolf, R. T. W., & Goldsmith, G. R. (2019). Seasonal origins of soil water used by trees. Hydrology and Earth System Sciences, 23(2), 1199–1210. https://doi.org/10.5194/hess-23-1199-2019
- Andrews, S. F., Flanagan, L. B., Sharp, E. J., & Cai, T. (2012). Variation in water potential, hydraulic characteristics, and water source use in montane Douglas-fir and lodgepole pine trees in southwestern Alberta and consequences for seasonal changes in photosynthetic capacity. *Tree Physiology*, 32(2), 146–160. https://doi.org/10.1093/treephys/tpr136
- Balugani, E., Lubczynski, M. W., Reyes-Acosta, L., van der Tol, C., Francés, A. P., & Metselaar, K. (2017). Groundwater and unsaturated zone evaporation and transpiration in a semi-arid open woodland. *Journal of Hydrology*, 547, 54–66. https://doi.org/10.1016/j. jhydrol.2017.01.042
- Barbeta, A., & Peñuelas, J. (2017). Relative contribution of groundwater to plant transpiration estimated with stable isotopes. *Scientific Reports*, 7(1), 10580. https://doi.org/10.1038/s41598-017-09643-x
- Beerling, D. J., & Berner, R. A. (2005). Feedbacks and the coevolution of plants and atmospheric CO2. Proceedings of the National Academy of Sciences, 102(5), 1302–1305. https://doi.org/10.1073/pnas.0408724102
- Beerling, D. J., Chaloner, W. G., Woodward, F. I., Algeo, T. J., & Scheckler, S. E. (1998). Terrestrial-marine teleconnections in the Devonian: Links between the evolution of land plants, weathering processes, and marine anoxic events. *Philosophical Transactions of the Royal Society of London. Series B: Biological Sciences*, 353(1365), 113–130. https://doi.org/10.1098/rstb.1998.0195
- Berkelhammer, M., Still, C. J., Ritter, F., Winnick, M., Anderson, L., Carroll, R., et al. (2020). Persistence and plasticity in conifer water-use strategies. Journal of Geophysical Research: Biogeosciences, 125(2), e2018JG004845. https://doi.org/10.1029/2018JG004845
- Blomberg, S. P., Garland, T., & Ives, A. R. (2003). Testing for phylogenetic signal in comparative data: Behavioral traits are more labile. *Evolution*, 57(4), 717–745. https://doi.org/10.1111/j.0014-3820.2003.tb00285.x
- Bowling, D. R., Schulze, E. S., & Hall, S. J. (2017). Revisiting streamside trees that do not use stream water: Can the two water worlds hypothesis and snowpack isotopic effects explain a missing water source? *Ecohydrology*, 10(1), e1771. https://doi.org/10.1002/eco.1771
- Brantley, S. L., Eissenstat, D. M., Marshall, J. A., Godsey, S. E., Balogh-Brunstad, Z., Karwan, D. L., et al. (2017). Reviews and syntheses: On the roles trees play in building and plumbing the critical zone. *Biogeosciences*, 14(22), 5115–5142. https://doi.org/10.5194/bg-14-5115-2017
- Brum, M., Vadeboncoeur, M. A., Ivanov, V., Asbjornsen, H., Saleska, S., Alves, L. F., et al. (2019). Hydrological niche segregation defines forest structure and drought tolerance strategies in a seasonal Amazon forest. *Journal of Ecology*, 107(1), 318–333. https://doi. org/10.1111/1365-2745.13022
- Cabal, C., Martínez-García, R., de Castro Aguilar, A., Valladares, F., & Pacala, S. W. (2020). The exploitative segregation of plant roots. *Science*, 370(6521), 1197–1199. https://doi.org/10.1126/science.aba9877
- Canadell, J., Jackson, R. B., Ehleringer, J. B., Mooney, H. A., Sala, O. E., & Schulze, E.-D. (1996). Maximum rooting depth of vegetation types at the global scale. *Oecologia*, 108(4), 583–595. https://doi.org/10.1007/BF00329030
- Chen, W., Zeng, H., Eissenstat, D. M., & Guo, D. (2013). Variation of first-order root traits across climatic gradients and evolutionary trends in geological time. *Global Ecology and Biogeography*, 22(7), 846–856. https://doi.org/10.1111/geb.12048
- Chen, Y., Helliker, B. R., Tang, X., Li, F., Zhou, Y., & Song, X. (2020). Stem water cryogenic extraction biases estimation in deuterium isotope composition of plant source water. *Proceedings of the National Academy of Sciences*, 117, 33345–33350. https://doi.org/10.1073/ pnas.2014422117
- Comas, L. H., & Eissenstat, D. M. (2009). Patterns in root trait variation among 25 co-existing North American forest species. *New Phytologist*, *182*(4), 919–928. https://doi.org/10.1111/j.1469-8137.2009.02799.x
- Debastiani, V. J., Bastazini, V. A. G., & Pillar, V. D. (2021). Using phylogenetic information to impute missing functional trait values in ecological databases. *Ecological Informatics*, 63, 101315. https://doi.org/10.1016/j.ecoinf.2021.101315
- De Deurwaerder, H., Hervé-Fernández, P., Stahl, C., Burban, B., Petronelli, P., Hoffman, B., et al. (2018). Liana and tree below-ground water competition—Evidence for water resource partitioning during the dry season. *Tree Physiology*, *38*(7), 1071–1083. https://doi.org/10.1093/treephys/tpy002
- Evaristo, J., & McDonnell, J. J. (2017). Prevalence and magnitude of groundwater use by vegetation: A global stable isotope meta-analysis. *Scientific Reports*, 7(1), 44110. https://doi.org/10.1038/srep44110
- Fan, Y. (2015). Groundwater in the Earth's critical zone: Relevance to large-scale patterns and processes. *Water Resources Research*, 51(5), 3052–3069. https://doi.org/10.1002/2015WR017037
- Fan, Y., Clark, M., Lawrence, D. M., Swenson, S., Band, L. E., Brantley, S. L., et al. (2019). Hillslope hydrology in global change research and Earth system modeling. *Water Resources Research*, 55(2), 1737–1772. https://doi.org/10.1029/2018WR023903
- Fan, Y., Miguez-Macho, G., Jobbágy, E. G., Jackson, R. B., & Otero-Casal, C. (2017). Hydrologic regulation of plant rooting depth. Proceedings of the National Academy of Sciences, 114(40), 10572–10577. https://doi.org/10.1073/pnas.1712381114
- Ferguson, I. M., Jefferson, J. L., Maxwell, R. M., & Kollet, S. J. (2016). Effects of root water uptake formulation on simulated water and energy budgets at local and basin scales. *Environmental Earth Sciences*, 75(4), 316. https://doi.org/10.1007/s12665-015-5041-z
- Fritz, S. A., & Purvis, A. (2010). Selectivity in mammalian extinction risk and threat types: A new measure of phylogenetic signal strength in binary traits. *Conservation Biology*, 24(4), 1042–1051. https://doi.org/10.1111/j.1523-1739.2010.01455.x
- Gaines, K. P., Stanley, J. W., Meinzer, F. C., McCulloh, K. A., Woodruff, D. R., Chen, W., et al. (2016). Reliance on shallow soil water in a mixed-hardwood forest in central Pennsylvania. *Tree Physiology*, 36(4), 444–458. https://doi.org/10.1093/treephys/tpv113
- Good, S. P., Noone, D., & Bowen, G. (2015). Hydrologic connectivity constrains partitioning of global terrestrial water fluxes. *Science*, 349(6244), 175–177. https://doi.org/10.1126/science.aaa5931
- Grossiord, C. (2020). Having the right neighbors: How tree species diversity modulates drought impacts on forests. *New Phytologist*, 228(1), 42–49. https://doi.org/10.1111/nph.15667

- Grossiord, C., Gessler, A., Granier, A., Berger, S., Bréchet, C., Hentschel, R., et al. (2014). Impact of interspecific interactions on the soil water uptake depth in a young temperate mixed species plantation. *Journal of Hydrology*, 519, 3511–3519. https://doi.org/10.1016/j. jhydrol.2014.11.011
- Grossiord, C., Sevanto, S., Dawson, T. E., Adams, H. D., Collins, A. D., Dickman, L. T., et al. (2017). Warming combined with more extreme precipitation regimes modifies the water sources used by trees. New Phytologist, 213(2), 584–596. https://doi.org/10.1111/nph.14192
- Guénard, G., Legendre, P., & Peres-Neto, P. (2013). Phylogenetic eigenvector maps: A framework to model and predict species traits. *Methods in Ecology and Evolution*, 4(12), 1120–1131. https://doi.org/10.1111/2041-210X.12111
- Hafner, B. D., Tomasella, M., Häberle, K.-H., Goebel, M., Matyssek, R., & Grams, T. E. E. (2017). Hydraulic redistribution under moderate drought among English oak, European beech and Norway spruce determined by deuterium isotope labeling in a split-root experiment. *Tree Physiology*, 37(7), 950–960. https://doi.org/10.1093/treephys/tpx050
- Hodge, A. (2004). The plastic plant: Root responses to heterogeneous supplies of nutrients. *New Phytologist*, *162*(1), 9–24. https://doi.org/10.1111/j.1469-8137.2004.01015.x
- Hupet, F., Lambot, S., Feddes, R. A., Dam, J. C. v., & Vanclooster, M. (2003). Estimation of root water uptake parameters by inverse modeling with soil water content data. Water Resources Research, 39(11), 1312. https://doi.org/10.1029/2003WR002046
- Ibarra, D. E., Rugenstein, J. K. C., Bachan, A., Baresch, A., Lau, K. V., Thomas, D. L., et al. (2019). Modeling the consequences of land plant evolution on silicate weathering. *American Journal of Science*, 319(1), 1–43. https://doi.org/10.2475/01.2019.01
- Iversen, C. M., McCormack, M. L., Powell, A. S., Blackwood, C. B., Freschet, G. T., Kattge, J., et al. (2017). A global fine-root ecology database to address below-ground challenges in plant ecology. New Phytologist, 215(1), 15–26. https://doi.org/10.1111/nph.14486
- Jin, Y., & Qian, H. (2019). V. PhyloMaker: An R package that can generate very large phylogenies for vascular plants. *Ecography*, 42(8), 1353–1359. https://doi.org/10.1111/ecog.04434
- Kalwij, J. M. (2012). Review of "The Plant List, a working list of all plant species". Journal of Vegetation Science, 23(5), 998–1002. https:// doi.org/10.1111/j.1654-1103.2012.01407.x
- Kennedy, D., Swenson, S., Oleson, K. W., Lawrence, D. M., Fisher, R., da Costa, A. C. L., & Gentine, P. (2019). Implementing plant hydraulics in the Community Land Model, version 5. Journal of Advances in Modeling Earth Systems, 11(2), 485–513. https://doi. org/10.1029/2018MS001500
- Kitajima, K., Anderson, K. E., & Allen, M. F. (2010). Effect of soil temperature and soil water content on fine root turnover rate in a California mixed conifer ecosystem. Journal of Geophysical Research, 115(G4). https://doi.org/10.1029/2009JG001210
- Kleidon, A., & Heimann, M. (1998). A method of determining rooting depth from a terrestrial biosphere model and its impacts on the global water and carbon cycle. Global Change Biology, 4(3), 275–286. https://doi.org/10.1046/j.1365-2486.1998.00152.x
- Klein, T. (2014). The variability of stomatal sensitivity to leaf water potential across tree species indicates a continuum between isohydric and anisohydric behaviors. *Functional Ecology*, 28(6), 1313–1320. https://doi.org/10.1111/1365-2435.12289
- Knighton, J., Conneely, J., & Walter, M. T. (2019). Possible increases in flood frequency due to the loss of eastern hemlock in the northeastern United States: Observational insights and predicted impacts. *Water Resources Research*, 55(7), 5342–5359. https://doi. org/10.1029/2018WR024395
- Knighton, J., Kuppel, S., Smith, A., Soulsby, C., Sprenger, M., & Tetzlaff, D. (2020). Using isotopes to incorporate tree water storage and mixing dynamics into a distributed ecohydrologic modeling framework. *Ecohydrology*, 13(3), e2201. https://doi.org/10.1002/eco.2201
- Knighton, J., Singh, K., & Evaristo, J. (2020). Understanding catchment-scale forest root water uptake strategies across the continental United States through inverse ecohydrological modeling. *Geophysical Research Letters*, 47(1), e2019GL085937. https://doi. org/10.1029/2019GL085937
- Knighton, J., Souter-Kline, V., Volkmann, T., Troch, P. A., Kim, M., Harman, C. J., et al. (2019). Seasonal and topographic variations in ecohydrological separation within a small, temperate, snow-influenced catchment. *Water Resources Research*, 55(8), 6417–6435. https:// doi.org/10.1029/2019WR025174
- Kulmatiski, A., Beard, K. H., Holdrege, M. C., & February, E. C. (2020). Small differences in root distributions allow resource niche partitioning. *Ecology and Evolution*, 10(18), 9776–9787. https://doi.org/10.1002/ece3.6612
- Link, P., Simonin, K., Maness, H., Oshun, J., Dawson, T., & Fung, I. (2014). Species differences in the seasonality of evergreen tree transpiration in a mediterranean climate: Analysis of multiyear, half-hourly sap flow observations. *Water Resources Research*, *50*(3), 1869–1894. https://doi.org/10.1002/2013WR014023
- Liu, Z., Yu, X., & Jia, G. (2019). Water uptake by coniferous and broad-leaved forest in a rocky mountainous area of northern China. Agricultural and Forest Meteorology, 265, 381–389. https://doi.org/10.1016/j.agrformet.2018.11.036
- Mackay, D. S., Savoy, P. R., Grossiord, C., Tai, X., Pleban, J. R., Wang, D. R., et al. (2020). Conifers depend on established roots during drought: Results from a coupled model of carbon allocation and hydraulics. *New Phytologist*, 225(2), 679–692. https://doi.org/10.1111/ nph.16043
- Magh, R.-K., Eiferle, C., Burzlaff, T., Dannenmann, M., Rennenberg, H., & Dubbert, M. (2020). Competition for water rather than facilitation in mixed beech-fir forests after drying-wetting cycle. Journal of Hydrology, 587, 124944. https://doi.org/10.1016/j.jhydrol.2020.124944
- McCormack, M. L., Kaproth, M. A., Cavender-Bares, J., Carlson, E., Hipp, A. L., Han, Y., & Kennedy, P. G. (2020). Climate and phylogenetic history structure morphological and architectural trait variation among fine-root orders. *New Phytologist*, 228(6), 1824–1834. https://doi.org/10.1111/nph.16804
- Miller, G. R., Chen, X., Rubin, Y., Ma, S., & Baldocchi, D. D. (2010). Groundwater uptake by woody vegetation in a semiarid oak savanna. *Water Resources Research*, 46(10), W10503. https://doi.org/10.1029/2009WR008902
- Molina-Venegas, R., & Rodríguez, M. Á. (2017). Revisiting phylogenetic signal; strong or negligible impacts of polytomies and branch length information? *BMC Evolutionary Biology*, 17. https://doi.org/10.1186/s12862-017-0898-y
- Münkemüller, T., Lavergne, S., Bzeznik, B., Dray, S., Jombart, T., Schiffers, K., & Thuiller, W. (2012). How to measure and test phylogenetic signal. *Methods in Ecology and Evolution*, *3*(4), 743–756. https://doi.org/10.1111/j.2041-210X.2012.00196.x
- Pagel, M. (1999). Inferring the historical patterns of biological evolution. Nature, 401(6756), 877-884. https://doi.org/10.1038/44766
  - Penone, C., Davidson, A. D., Shoemaker, K. T., Marco, M. D., Rondinini, C., Brooks, T. M., et al. (2014). Imputation of missing data in life-history trait data sets: Which approach performs the best? *Methods in Ecology and Evolution*, 5(9), 961–970. https://doi. org/10.1111/2041-210X.12232
  - Pierret, A., Maeght, J.-L., Clément, C., Montoroi, J.-P., Hartmann, C., & Gonkhamdee, S. (2016). Understanding deep roots and their functions in ecosystems: An advocacy for more unconventional research. Annals of Botany, 118(4), 621–635. https://doi.org/10.1093/aob/mcw130
  - Rodríguez-Robles, U., Arredondo, J. T., Huber-Sannwald, E., Yépez, E. A., & Ramos-Leal, J. A. (2020). Coupled plant traits adapted to wetting/drying cycles of substrates co-define niche multidimensionality. *Plant, Cell and Environment, 43*(10), 2394–2408. https://doi.org/10.1111/pce.13837

- Schenk, H. J., & Jackson, R. B. (2005). Mapping the global distribution of deep roots in relation to climate and soil characteristics. *Geoderma*, 126(1), 129–140. https://doi.org/10.1016/j.geoderma.2004.11.018
- Silvertown, J., Araya, Y., & Gowing, D. (2015). Hydrological niches in terrestrial plant communities: A review. Journal of Ecology, 103(1), 93–108. https://doi.org/10.1111/1365-2745.12332
- Sulis, M., Couvreur, V., Keune, J., Cai, G., Trebs, I., Junk, J., et al. (2019). Incorporating a root water uptake model based on the hydraulic architecture approach in terrestrial systems simulations. *Agricultural and Forest Meteorology*, 269(270), 28–45. https://doi.org/10.1016/j. agrformet.2019.01.034
- Sun, L., Yang, L., Chen, L., Zhao, F., & Li, S. (2018). Hydraulic redistribution and its contribution to water retention during short-term drought in the summer rainy season in a humid area. *Journal of Hydrology*, 566, 377–385. https://doi.org/10.1016/j.jhydrol.2018.09.032 Tetzlaff, D., Buttle, J., Carey, S. K., Kohn, M. J., Laudon, H., McNamara, J. P., et al. (2021). Stable isotopes of water reveal differences in
- Perzian, D., Butte, J., early, S. K., Konn, M. J., Eaudon, M., Mervanara, J. T., et al. (2021). Stable isotopes of water reveal unreferences in plant-soil water relationships across northern environments. *Hydrological Processes*, 35(1), e14023. https://doi.org/10.1002/hyp.14023 Trugman, A. T., Anderegg, L. D. L., Anderegg, W. R. L., Das, A. J., & Stephenson, N. L. (2021). Why is tree drought mortality so hard to
- predict? Trends in Ecology & Evolution, 36(6), 520–532. https://doi.org/10.1016/j.tree.2021.02.001 Urli, M., Porté, A. J., Cochard, H., Guengant, Y., Burlett, R., & Delzon, S. (2013). Xylem embolism threshold for catastrophic hydraulic
- failure in angiosperm trees. *Tree Physiology*, *33*(7), 672–683. https://doi.org/10.1093/treephys/tpt030
- Valverde-Barrantes, O. J., Freschet, G. T., Roumet, C., & Blackwood, C. B. (2017). A worldview of root traits: The influence of ancestry, growth form, climate and mycorrhizal association on the functional trait variation of fine-root tissues in seed plants. *New Phytologist*, 215(4), 1562–1573. https://doi.org/10.1111/nph.14571
- Volkmann, T. H. M., Haberer, K., Gessler, A., & Weiler, M. (2016). High-resolution isotope measurements resolve rapid ecohydrological dynamics at the soil-plant interface. *New Phytologist*, 210(3), 839–849. https://doi.org/10.1111/nph.13868
- Wang, C., McCormack, M. L., Guo, D., & Li, J. (2019). Global meta-analysis reveals different patterns of root tip adjustments by angiosperm and gymnosperm trees in response to environmental gradients. *Journal of Biogeography*, 46(1), 123–133. https://doi.org/10.1111/ jbi.13472
- Wang, P., Niu, G.-Y., Fang, Y.-H., Wu, R.-J., Yu, J.-J., Yuan, G.-F., et al. (2018). Implementing dynamic root optimization in Noah-MP for simulating phreatophytic root water uptake. Water Resources Research, 54(3), 1560–1575. https://doi.org/10.1002/2017WR021061
- Warren, J. M., Hanson, P. J., Iversen, C. M., Kumar, J., Walker, A. P., & Wullschleger, S. D. (2015). Root structural and functional dynamics in terrestrial biosphere models—Evaluation and recommendations. *New Phytologist*, 205(1), 59–78. https://doi.org/10.1111/nph.13034
- Warton, D. I., Wright, I. J., Falster, D. S., & Westoby, M. (2006). Bivariate line-fitting methods for allometry. *Biological Reviews*, 81(2), 259–291. https://doi.org/10.1017/S1464793106007007
- Yang, Y., Guan, H., Hutson, J. L., Wang, H., Ewenz, C., Shang, S., & Simmons, C. T. (2013). Examination and parameterization of the root water uptake model from stem water potential and sap flow measurements. *Hydrological Processes*, 27(20), 2857–2863. https://doi. org/10.1002/hyp.9406